Author's response to reviews

Title: Children's active play: self-reported motivators, barriers and facilitators.

Authors:

Rowan Brockman (Rowan.Brockman@bristol.ac.uk)
Russell Jago (Russ.Jago@bristol.ac.uk)
Kenneth R Fox (K.R.Fox@bristol.ac.uk)

Version: 5 Date: 9 June 2011

Author's response to reviews: see over
Dear Editorial Team

Thank you for accepting our manuscript, ‘Children’s active play: self-reported motivators, barriers and facilitators’ for publication in BMC Public Health.

Please find below a point-by-point descriptions of the formatting changes requested.

Major Compulsory Revisions

1. Author’s contributions: Please indicate that all authors read and approved the final manuscript.

   This has now been added.

Minor Revisions

1. Author list: please remove authors’ titles.

   This has now been amended.

2. Highlighting/tracking: Please remove all highlighting and tracking from the manuscript.

   This has now been removed.

3. Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions.

   This has been checked – no corrections required.

END